Author's response to reviews

Title: Maternal exposure to polychlorinated biphenyls and the secondary sex ratio: an occupational cohort study

Authors:

Carissa M Rocheleau (CRocheleau@cdc.gov)
Stephen J Bertke (SBerke@cdc.gov)
James A Deddens (JDeddens@cdc.gov)
Avima M Ruder (ARuder@cdc.gov)
Christina C Lawson (CLawson@cdc.gov)
Martha A Waters (MWaters@cdc.gov)
Nancy B Hopf (Nancy.Hopf@hospvd.ch)
Margaret A Riggs (MRiggs@cdc.gov)
Elizabeth A Whelan (EWhelan@cdc.gov)

Version: 4 Date: 24 February 2011

Author’s response to reviews:

Editorial corrections have been made in this version, as requested.

1. Affiliations are written out in full, including city and state. Some of the department names are lengthy, but each numbered affiliation is a single affiliation.

2. We made sure that the abstract text in the online submission and in this document are identical.

3. Tables have been edited so that vertical lines are not visible.

4. Where appropriate, we use 'concentration' instead of 'level' (e.g. 'PCB serum concentration).

5. In table 1 the mis-spelling of primipara is corrected.